Author's response to reviews

Title: Thrombosed Traumatic Aneurysm of the Occipital Artery: Case Report and Review of the Literature

Authors:

Vikas Y Rao MD (vikasr@bcm.edu)
Steven W Hwang MD (stevenhwang@hotmail.com)
Adekunle M Adesina MD PhD (amadesin@texaschildrens.org)
Andrew Jea MD (ajea@bcm.edu)

Version: 2 Date: 23 February 2012

Author's response to reviews: see over
January 4, 2012

Dear Editor,

We are submitting for your consideration our paper entitled: “Thrombosed Traumatic Aneurysm of the Occipital Artery: Case Report and Review of the Literature.” Aneurysms of the occipital artery are rare; most are usually pseudoaneurysms. Currently, most cases in the literature report post-traumatic pseudoaneurysms of the occipital artery. We present a case of a true aneurysm of the occipital artery after minor head trauma. This is an original manuscript, and no part of it has been published elsewhere. Our suggested peer reviewers are:

Dr. Sanjiv Bhatia (sbhatia@med.miami.edu), Dr. James Rutka (james.rutka@sickkids.ca), and Dr. James Drake (james.drake@sickkids.ca). Thank you for taking the time to consider our manuscript. We look forward to hearing from you.

Andrew Jea MD

Corresponding Author